Author’s response to reviews

Title: Ethical Challenges in Global Health-related Stigma Research

Authors:

Joseph Millum (millumj@cc.nih.gov)
Megan Campbell (info@drmegancampbell.co.za)
Florence Luna (florlunaflaco@gmail.com)
Arianne Malekzadeh (arianne.malekzadeh@nih.gov)
Quarraisha Karim (Quarraisha.AbdoolKarim@caprisa.org)

Version: 1 Date: 10 Jan 2019

Author’s response to reviews:

10 January 2019

Dear Dr. Lopez

Many thanks for the positive response and the further comments! I am writing to re-submit our manuscript to BMC Medicine. The title of the manuscript is ‘Ethical Challenges in Global Health-related Stigma Research.’

I have uploaded two versions of the revised manuscript – one clean and one with our changes tracked. I have also uploaded a point by point response to the reviewer comments that explains the changes we have made.

We thank the reviewers for their further comments and think that the changes we’ve made in response have further improved the manuscript. Please let me know if you need anything else from me.

Sincerely

Joseph Millum, Ph.D., M.Sc.